Author's response to reviews

Title: Bone mineral density, body mass index and cigarette smoking among Iranian women: Implications for prevention

Authors:

Azam Baheiraei (a.baheiraei@garvan.org.au)
Nicholas A Pocock (n.pocock@unsw.edu.au)
John A Eisman (j.eisman@garvan.org.au)
Nguyen D Nguyen (n.nguyen@garvan.org.au)
Tuan V Nguyen (t.nguyen@garvan.org.au)

Version: 6 Date: 14 June 2005

Author's response to reviews: see over
June 14 2005

Dr. Elizabeth C. Moylan
Assistant Editor
BioMed Central
Middlesex House
32-42 Cleveland Street
London
W1T 4LB
UK

Dear Dr Moylan,

MS: 7837286305676117
Bone mineral density, body mass index and cigarette smoking among Iranian women: Implications for prevention

Thank you for your e-mail dated 09/06/2005 together with the Reviewers’ comments on the above manuscript. We have carefully checked the manuscript for any errors. We have changed the formatting as requested and only a few words to the text. We hope that our revised manuscript is satisfactory to you and suitable for publication in the Journal.

Yours sincerely,

Tuan V Nguyen
Formatting changes:

1. We have used the sentence case in the title.
2. The figures have been cropped to minimise white spaces around the images.

Text changes:

1. We have removed the sentence “currently living in Sydney” in the manuscript (page 4 Paragraph 2 line 4).
2. We have added “or not employed” and “the” has been corrected to “their” in page 7 line 7.
3. The expression “of residence” has now been added in page 7 line 8.
4. We have replaced the word “is” with “was” (page 10 line 3).
5. We have corrected “comparable” to read “generalizable” in the Discussion section (page 11 paragraph 2 line 6).